

The Green Side of Industry: The Drivers and the Impacts of ECO-Innovations in Brazil

Sperotto, Fernanda Q. and Tartaruga, Iván G. P.

Centre of Studies in Geography and Spatial Planning (CEGOT), Faculty of Arts and Humanities, University of Porto, Centre of Studies in Geography and Spatial Planning (CEGOT), Faculty of Arts and Humanities, University of Porto

19 July 2021

Online at https://mpra.ub.uni-muenchen.de/109149/ MPRA Paper No. 109149, posted 20 Jul 2022 09:53 UTC





Article The Green Side of Industry: The Drivers and the Impacts of ECO-Innovations in Brazil⁺

Fernanda Q. Sperotto ^{1,2,*} and Iván G. P. Tartaruga ^{1,2,*}

- ¹ Centre of Studies in Geography and Spatial Planning (CEGOT), 4150-564 Porto, Portugal
- ² Faculty of Arts and Humanities, University of Porto, Via Panorâmica s/n, 4150-564 Porto, Portugal
- * Correspondence: fsperotto@letras.up.pt (F.Q.S.); itartaruga@letras.up.pt (I.G.P.T.)
- + This is a modified version of a conference document. In Proceedings of the V Industrial Economy and Innovation Meeting (ENEI): Innovation, Sustainability and Pandemic, Belo Horizonte, Brazil, 10–14 May 2021.

Abstract: This study aimed to provide an overview of eco-innovations in the Brazilian industry. To address this issue, we analyzed specific data of eco-innovative companies. In addition, we applied the cluster heatmap technique, which allowed us to analyze the different drivers and impacts of eco-innovations in different sectors. According to the results, companies that stated that innovation made it possible to reduce their environmental impact represent a third of all innovators. Moreover, they are companies that have shown greater effort to innovate and greater susceptibility to the benefits and obstacles of innovation. Furthermore, the eco-innovation strategy is mainly driven by market factors, such as reputation and codes of good practice. The impacts are mainly associated with the use of more widespread and less complex technologies, such as recycling. In addition to these results, the study considers some alternatives to guide the innovation policy, especially related to eco-innovations in semi-peripherical countries.



Citation: Sperotto, F.Q.; Tartaruga, I.G.P. The Green Side of Industry: The Drivers and the Impacts of ECO-Innovations in Brazil . *Sustainability* **2021**, *13*, 8065. https://doi.org/10.3390/su13148065

Academic Editor: Iwona Zdonek

Received: 23 June 2021 Accepted: 14 July 2021 Published: 19 July 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Keywords: sustainability; green technologies; environmental innovation; industry; Brazil

1. Introduction

The climate change scenario requires the application of a set of actions capable of tackling ecological, social, and economic problems. Harmonizing the economic, social, and environmental dimensions is at the core of the 2030 Agenda and the Sustainable Development Goals (SDGs).

The perspective that will allow the development of a sustainable technological model is mainly connected with two SDGs. The first, SDG 7 (affordable and clean energy), aims to ensure universal access to affordable, reliable, and modern energy services. Therefore, its purposes include: substantially increasing the share of renewable energy in the global energy mix; improving international cooperation to facilitate access to clean energy research and technology; and widening infrastructure and upgrading technology for supplying modern and sustainable energy services for all nations, especially in developing countries. The second, SDG 9 (industry, innovation, and infrastructure), aims to build resilient infrastructures, to promote inclusive and sustainable industrialization and to encourage innovations. Its included targets aim to: modernize infrastructures and to retrofit industries for strengthening sustainability, with increased resource-use efficiency and greater adoption of clean and environmentally sound technologies and industrial process; intensify scientific research, for improving the technological capabilities of industrial sectors; and encouraging innovation and increasing the number of workers and R&D resources (public and private), particularly in developing countries [1].

The preview targets in two SDGs are interconnected to the purposes of eco-innovations (EIs). The EIs follow practically the same features as general innovations—products, processes, and organizational techniques, which can be new to a market or company—but with the differential of reducing the environmental risks throughout its life cycle [2–4].

In discussions about technological changes, there are high expectations regarding the potential of green technologies [5–7] and the competitive advantages of EIs [8]. Investment targeting renewable energies, biotechnologies, and more technically and ecologically efficient products and processes is key to a new technological model [9–11]. In this respect, it is worthwhile to note that the principal economic policies today are closely tied to sustainable development, such as the European Green Deal in the European Union or the Green New Deal in the United States. In line with this paradigm shift in technology, development opportunities also arise for developing countries, specifically, in Latin America [12,13].

At the same time, there are challenges, especially for developing countries [14]. In the case of Brazil, the transition to a green economy and the encouragement of EIs has outlines that are well delineated to that of a developing country, requiring, above all, technological, financial, and institutional efforts. The ideal path presupposes that the country can promote structural change that increases productivity in a socially inclusive [15] and environmentally sustainable manner. However, in fact, this nation has provided a dual response to this situation in terms of government and public policy in innovation [16]. On the one hand, Brazilian innovation systems have some strengths built on the existence of all the components of a significant innovation system, for example, educational, research and regulation subsystems are important for our purposes here, the strategic natural resources, such as land, mineral and water supplies as well as biodiversity from land and maritime biomes. On the other hand, there are key system weaknesses, such as the frequent negative macroeconomic interventions in innovative productive activities and, perhaps the most critical, the absence of a consistent long-term vision or agenda, i.e., fragmentation and discontinuity in public policies and institutional configurations, that give direction and coherence to scientific research and public and private innovation efforts.

In this context, an important question is how developing countries, such as Brazil, have responded to the transition to a more sustainable development model. To illuminate this issue, the analysis combined elements of innovation, production and environmental protection from a sectorial perspective. The study examined specific EI data—drivers and effects—in Brazilian companies, considering two delimitations. The first concerns the identification of the group of eco-innovative companies, through the application of a specific selection criterion, directing the analysis to innovative companies that observed an environmental benefit from their generated innovation. The second delimitation relates to the sectorial profile, namely, the extractive and manufacturing sectors. The database is from the latest edition of Pintec (Pesquisa de Inovação), 2015–2017, which included specific issues about EIs for the first time.

Thus, the objective of this paper was to provide an overview of eco-innovations in Brazilian industry. Therefore, a comparative profile of two groups of companies (ecoinnovators and innovators) was elaborated and, in the case of eco-innovators, the drivers and impacts of eco-innovations were analyzed. We believe that analyzing eco-innovative companies, especially in developing countries, is extremely relevant for challenges of the 2030 Agenda. The panorama of EIs in Brazilian extractive and manufacturing sectors indicated that the eco-innovative companies represent just over a third of the innovators. In general, these companies show a greater effort to innovate when compared to other innovators [17,18]. Associated with this greater commitment, eco-innovators demonstrate a greater perception of the benefits generated by innovation. Likewise, they also exhibit greater sensitivity to its obstacles. The analysis of drivers and impacts allowed us to enrich this profile, highlighting some central elements of environmental innovations, such as the fact that EIs in the Brazilian extractive and manufacturing activities are mainly induced by market factors, and that environmental regulations are not part of the most relevant determinants. Moreover, the most recurrent effects are related to the use of more widespread and less complex technologies, such as the recycling of materials, waste, and wastewater [19].

The article is structured as follows: Section 2 is dedicated to the theme of EI, with an emphasis on drivers and the effects that environmental innovation potentially generates.

Section 3 describes the methodology, in particular, the selection criteria, of the companies and the elaboration of the graphic elements of analysis. Section 4 focuses on results, with special attention to information from Pintec sustainability and the EI set. Section 5 discusses the main findings. Finally, Section 6 concludes by indicating possibilities to guide innovation policy, especially regarding eco-innovations in semi-peripheral countries.

2. Literature Review

For a long time, environmental protection actions were seen as an impediment to the economy, reinforcing the existence of a permanent trade-off between these two dimensions. Although the attachment to this idea is still manifested, technological advances and institutional conduct strategies (government, companies, civil society, NGOs, etc.) already allow us to make viable alternative and sustainable models of production and consumption for which EIs and green technologies are crucial.

In this direction, the proposition of Porter and Line [20] was a precursor in demonstrating that pollution was equivalent to inefficiency. When analyzing some sectors that were strongly influenced by environmental regulations, they found that the adoption of innovations and other factors that promote competitiveness allowed companies to reduce costs with environmental actions. Among other aspects, environmental standards were able to indicate inefficiencies, encouraging new solutions; sensitize companies about their production processes; reduce uncertainty about investments for environmental improvement; and pressure companies to eco-innovate [20].

Technological innovation policies play a primary role for the purposes of sustainable development. Therefore, the substitution of unsustainable production methods and consumption patterns by innovations capable of mitigating impacts on the environment is part of a broader context of technological evolution [21]. Thus, it is especially important that innovation and sustainability are developed together, as the needs foreseen in the meaning of sustainable development are dynamic and, consequently, susceptible to changes in the current context [22].

The concept of EI adopted in this analysis is from the report *Measuring ECO-innovation*: "[...] the production, assimilation or exploitation of a product, production process, service or management or business method that is novel to the organization (developing or adopting it) and which results, throughout its life cycle, in a reduction of environmental risk, pollution and other negative impacts of resources use (including energy use) compared to relevant alternatives." [2] (p. 7).

Thus, a company can become an eco-innovator when it applies environmental technology, e.g., by developing equipment or processes that use inputs in a sustainable and efficient manner and/or reduces the effects of production on the most varied natural ecosystems. The manufacture of "green" products, the shared use of goods and the creation of environmental innovation systems, combining production and consumption to reduce impacts on the environment—such as organic products and renewable energies—are other examples. Another strategy is to adopt managerial or organizational methods, such as environmental management and certification systems, and prevention and cooperation programs, with other entities aimed at reducing the environmental impact [2,4].

A particular characteristic of EIs is the presence of the problem of double externality. As with in innovations in general, there are market failures due to knowledge externalities; in EIs, there are also failures attributed to the environmental benefits generated, which increases the risk, discouraging the development of EIs [23,24]. Therefore, the application of environmental standards is essential to encourage EIs and to reduce the double externality for companies that develop innovations.

As highlighted, our focus in this study was the impacts and drivers of EIs. In relation to the effects, or impacts, of EIs, these are commonly associated with the notion of performance. Studies that analyze impacts emphasize both the results directly perceived in the environment—for instance, the replacement or more efficient use of energy and inputs, and

the reduction in contamination in soil, water, and air—and others of economic character, such as financial performance, competitiveness, and market value [25–28].

In addition, there is a classification that links the impacts of EIs to a kind of innovation. The effects of product EIs refer to improving the recycling of products after use; reducing energy consumption; and reducing emissions in water, soil, and noise. The impact of process EIs corresponds, in addition to the last two products' effects, to the reduction in materials per product unit; the decrease in energy use per product unit; the decrease in CO_2 and other pollutants in the air; recycling waste, water or materials; and replacing hazardous substances [8].

The factors that induce companies to seek solutions to reduce their environmental impact, either reactively or proactively [25,27], are among the most analyzed aspects in the literature [26,29].

To analyze the drivers of EIs, we adopted the theory of environmental innovation [17], which includes both general aspects of innovations and particular aspects of EIs. Just as Horbach [8,17] and Triguero et al. [30], we categorized the determinants into three main groups: supply-side factors, demand-side factors, and institutional and political influences of an environmental characteristic. While the first two groups are widely known in innovation studies, the third—environmental and policy influences—is specific to EIs [31,32].

Hence, based on the factors listed by the theory of environmental innovation [17], it was possible to associate in each group of determinants the drivers present in the special survey by Pintec (Table 1). Pintec listed nine specific factors that contributed to the company's eco-innovation decision.

Determinants	Nature of Factors	Inducers (Pintec)				
Supply-side	Technology push Cost-savings	Government support High costs of energy, water, raw materials				
Demand-side	Market pull	Voluntary actions Codes of good practice Market demand Reputation				
Environmental and policy influences	Regulatory pull/push	Existing environmental standards Future environmental standards Public contract requirements				

Table 1. Determinants, nature of factors and inducers (Pintec) of environmental innovation.

Source: Adapted from [8,17,30].

The supply side factors are those related to the technological capabilities available for the company to innovate [33]. As with innovations in general, the factors correspond to the company's technological and organizational capabilities [17,34]; the cooperative relations with suppliers, research institutes, agencies and universities [8,30,32,35,36]; the access to external information and knowledge [35,36]; and the access to subsidies and fiscal incentives [31]. Other factors related to the supply-side are energy and material cost-saving strategies [8,34,37].

In turn, demand-side elements highlight market forces as the main determinants of technical change and, therefore, of stimulating (eco)innovations [33]. Among the main factors, the market pull is the possibility of expanding market share [8,17,38]; the increased market demand for green products and eco-label adoption strategies [15,39,40]; and the social awareness actions of the need for clean production [26,38], such as voluntary actions or codes of good practice, mainly resulting from pressure from stakeholders and consumers [22,31,34]. In particular, the institutional pressure can trigger a technological change within the company and therefore redirect R&D efforts for projects related to the environment [41].

A specificity of the determinants related to environmental regulation is the notion of regulatory pull/push. The empirical verification shows that the regulatory structure and environmental policy generates significant effects on EIs. This is due to the fact that these, unlike those in the fields of microelectronics or telecommunications, are not self-applicable; that is, the factors of market pull and technology push are not strong enough to develop them. Therefore, EIs need specific regulatory support, capable of inducing (pull) and boosting them (push) [23]. In this group of drivers, there are environmental standards and regulations, including the expectation of stricter future standards; environmental taxes and duties; and indirect instruments, such as the imposition of requirements for public procurement. It is worth mentioning that the effects of environmental regulation of the development of EIs is among the main topics in the analysis of environmental innovation drivers [8,18,20,22,31,37,42].

3. Material and Methods

The research was structured in two stages (Figure 1). The first focused on the selection criteria for eco-innovative companies. A significant part of quantitative studies on innovation uses surveys, such as the Community Innovation Survey (European Union) or Business R&D and Innovation Survey (United States). In particular, the surveys are a relevant base of information about drivers of EIs [8,15,43]. In the case of Brazil, the most well-known base of innovation statistics is Pintec, from the Instituto Brasileiro de Geografia e Estatística (IBGE). Pintec is a sample survey, conducted over a three-year period, and applied to companies with more than 10 employees belonging to the sectors of industry, electricity and gas, and selected services [44]. In addition to contemplating the unprecedented information from the last Pintec about the drivers and impacts of EIs, the objective of this study was to specifically analyze eco-innovative companies. To this end, questions related to the theme of innovation and the environment were identified in the general questions block. The question chosen to differentiate companies between innovators and eco-innovators was Nº 105, which asks about the effects of product and process innovations in reducing the impacts on the environment [44,45]. In this question, companies must manifest this effect, indicating one of four levels of importance: high, medium, low, and not relevant. The issue, as proposed in some studies [8,17,18,46,47], is a measure of environmental performance and, therefore, a proxy to differentiate eco-innovative companies from others.



Figure 1. Design of the stages of the study methodology.

Nevertheless, it is pertinent to make two brief considerations about the data. However, the survey also probes about the reduction in the consumption of other inputs (energy, water, and materials); this effect results much more often from cost saving, rather than from an initial guideline of environmental conduct [2,47]. Another point is the bias present in approaches on the environmental impact. The responses of companies may be subject to corporate rhetoric and, consequently, may be overestimated [48].

Based on the definition of the criteria for identifying eco-innovative companies, two databases were requested: one formed by innovative companies that declared "high" or "medium" importance in reducing the impact on the environment, hereinafter identified as ECO, and another composed of companies that answered "low" or "not relevant" relevance, identified as INO. The objective here was to draw a comparative profile between ECO and INO, based on a selection of variables from Pintec's general block, which were considered relevant for the purpose of the investigation. Data were derived from the extractive and manufacturing sectors, for the period 2015–2017.

The second stage involved the development of cluster heatmaps to examine information from the special EI set. The cluster heatmap statistical tool was developed computationally by Galili et al. [49], using the R language [50]. In simple terms, the method is applied to large data matrices, for viewing patterns and creating dendrograms for the definition of clusters.

The heatmap cluster method is based on the application of an algorithm, which provides a structure in a series of clusters, and clusters of clusters, of data examination. In other words, unlike other clustering methods, here the algorithm determines the number of clusters. These algorithms can follow different logics, the most common is hierarchical cluster analysis (HCA). This logic combines objects that are closest to each other until all objects are joined in one single cluster, which will be separated from other furthest objects. To analyze which drivers and impacts were more (or less) important for eco-innovative companies of different sectors, we applied the "complete linkage" method (for HCA) to calculate distances between aggregates (or dissimilarity between clusters). This method determines the distance (or dissimilarity) between two clusters as the distance between the furthest two objects from each cluster. Thus, the ordering method allows similar rows (or columns) to be placed together in the graph. The "complete linkage" method results in compact and homogeneous clusters [51], which we consider adequate for our study.

For the construction of these clusters, two questions from Pintec were selected: N°196, which asked companies to highlight the factors that contributed to their decision to introduce innovations that would generate environmental benefits, and N°105 (sub-items 1 to 5), which requested companies to indicate the relevance of the impact on the environment of the innovations implemented. For these sub-items, only "high" or "medium" graduations were considered.

The heatmap method is performed in three phases [49,51]: treatment of the data matrix; definition of similarity; and determination of the serialization of rows and columns matrix (dendrogram). The first concerns the organization of data. In this phase, two matrices of standardized data in "z" scores were composed by economic activity (two sections and 24 NACE divisions). As the number of companies in the sample is highly varied between activities, it was necessary that the data be standardized. Thus, the matrices were as follows: EI drivers, composed of 10 EI inducers and 26 economic activities, and EI impacts, formed by 5 observed effects and 26 economic activities (Figure 2). In the second, the type of similarity between the observations in each cell (row and column) was defined. In the third and last phases, the ranking of the data in the rows and columns was determined. Furthermore, the ordering of the data (scale) was carried out according to its distribution in the columns. Therefore, the colors and the order of the observations were distributed to highlight the sectors that had the highest values in each variable.

Industrial activities (2 sections + 24 divisions NACE)	Eco-innovation drivers matrix	Eco-innovation impact matrix
 Extractive (mining and quarrying) Manufacturing Man. food products Man. beverages Man. tobacco products Man. textiles Man. textiles Man. wearing apparel Man. wood and prod. wood and cork Man. opper and paper products Printing and reproduction of recorded media Man. coke and refined petroleum products Man. themicals and chemical products Man. other non-metallic mineral products Man. fabricated metal products Man. fabricated metal products Man. computer, electronic and optical products Man. machinery and equipment Man. other transport equipment Man. furniture Other manufacturing Repair and install. machinery and equipment 	 Voluntary actions (VACT) Meet the necessary requirements for the consolidation of public contracts (RQPC) Codes of good environmental practices in the sectors which it operates (COGP) Market demand (real or potential) for environmental innovation (MKTD) Availability of government support, subsidies or other government financial incentives for environmental innovation (GOVS) High costs or energy, water or raw materials (HCTS) Improve the company's reputation (REPT) Existing environmental standards or taxes on contamination (EEST) Environmental standards or taxes that may be introduced in the future (FEST) Others (OTRS) 	 Recycling of waste, wastewater or materials for sale or reuse (RECY) Reduced contamination of soil, water, noise or air (RCON) Reduction of the company's CO₂ footprint (CO₂ production) (RCO2) Total or partial replacement on energy of fossil fuel with renewable energy sources (RENG) Substitution, in whole or in part, of materials with others that are less pollutants or hazardous (SBSM)

Figure 2. Drivers and impacts of eco-innovation, and industrial activities to generate heatmap matrices.

4. Results

As highlighted in the concept of EI, environmental innovation occurs within the company and follows the characteristics of innovations in general, except for the positive effect it has on the environment. In this sense, according to the guidelines of the Oslo Manual [52] applied at Pintec [44], (eco)innovation corresponds to any product, process, or organizational form that is new to the company, even if it is known in the market. This section has two objectives: to draw a comparative profile between the ECO and INO companies, and to examine the drivers and impacts of EIs on ECOs.

4.1. Comparative Profile of ECO and INO Companies

The last edition of Pintec surveyed 39,329 innovative companies (all economic sectors covered by the survey); out of these, 13,365 (34.0%) said that innovation allowed them to reduce the environmental impact with "high" and "medium" importance. In extractive activity, there were 143 eco-innovative companies, while in manufacturing, there were 12,334 [45].

As shown in Table 2, the rates of innovation—total and sectorial—for ECOs were lower, as expected, since they represent a little more than a third of innovative companies. However, when analyzing the rate of innovative intensity (the ratio between the amount spent on innovation activities and the company's current net revenue), a slightly higher effort is observed in the ECOs, mainly in-house R&D activities [18,42,53]. As EIs represent a very dynamic field in which knowledge is rapidly changing, it is observed that R&D activities, as well as cooperation with research centers and universities, are more important compared to other innovations without environmental benefits [35,37,53–55]. Furthermore, as EIs, particularly products and processes, are characterized by a high degree of innovation, the cost of new equipment, software, and marketing activities is more relevant when compared to other innovations [8,54].

Sectors	Innovative Intensity Rate											
	Innovation Kate		Total		In-House R&	D Activities	- IN OI Companies					
	ECO	INO ECO INO		ECO	INO	ECO INO		Total				
Extractive	6.2	8.4	1.2	1.9	0.7	0.5	143	193	336			
Manufacturing	12.3	22.0	1.6	1.1	0.6	0.4	12,334	22,062	34,396			
Total ⁽¹⁾	11.4	22.2	1.8	1.6	0.9	0.5	13,365	25,964	39,329			

Table 2. Number, innovation rate, and innovative intensity rates of eco-innovative (ECO) and innovative (INO) companies, by selected economic sector, Brazil, 2015–2017.

Source: Own elaboration, based on IBGE [45]. Note: (1) Includes all sectors of the survey.

The degree of novelty of product and process innovation is an important measure, as it reflects the radical nature of innovation. While national rates refer to innovations for the domestic market already developed in the world market, world rates refer to the most substantial innovations, which are also new abroad (Figure 3).



Figure 3. Rate of product and process innovation in industrial sectors of eco-innovative (ECO) and innovative (INO) companies, according to the degree of novelty for market, Brazil, 2015–2017. Source: Own elaboration, based on IBGE [45].

When comparing these rates in groups of companies, there are interesting elements. In turn, in the world segment, although they present a higher index only in product innovation in the extractive sector, the difference between the two groups is smaller when compared to the national segment, indicating that the ECO and INO are practically at the same level of novelty. On the other hand, in manufacturing, the superiority of the INO in both markets is undeniable. In summary, the 2015–2017 portrait shows that Brazilian EIs generated a higher impact on extractive activities. This is positive, given that this sector is notoriously known for its high polluting potential and its ability to deplete non-renewable resources. Furthermore, incremental technologies predominate in Brazilian EIs, which has been observed in studies of environmental innovation in other countries [8].

The benefits observed with the innovations are relevant aspects for the discussion about the drivers and impacts of EIs, as they highlight a double stimulus effect: the inducing element reflects on good results that corroborate with the induction and encourage it. To address this issue, only companies that expressed "high" or "medium" importance were considered. In addition, these gains were grouped into three categories: competition (related to better market positioning), production (related to greater efficiency in the production process) and protection (linked to compliance with norms and standards). The



point that draws attention is the greater perception of ECO about the benefits generated by innovation (Figure 4). The only benefit in which INO had greater relevance was the opening of new markets in the extractive activities.

Figure 4. Benefits highlighted by eco-innovative (ECO) and innovative (INO) companies, in the extractive (**a**) and manufacturing (**b**) sectors, according to category, Brazil, 2015–2017. Source: Own elaboration, based on IBGE [44]. Note: data of companies that attributed "high" or "medium" importance to the benefit.

The main differences in favor of ECO were found in the categories of production and protection. In manufacturing (Figure 4b), the gains from the reduction in the consumption of inputs, energy, water and raw materials were the most relevant, and, in the protection category, both benefits from compliance with norms and standards were accentuated. In extractive sector (Figure 4a), the advantages of competition stood out, particularly the gains from the expansion and maintenance of the market share. In the case of the INOs, it is worth emphasizing that, despite manifesting benefits of reducing the consumption of water, energy, and materials, they declared that the innovation had a low or non-relevant impact on the environment. Here, the hypothesis is that the benefits of the reductions were, above all, the result of a cost saving strategy. Therefore, ECOs expressed the benefits of innovation more intensely—not only those that positively interfere in the environment (such as the decrease in the quantity of inputs), but also others capable of promoting them to more competitive environments. Although it may seem contradictory, eco-innovative companies appear to be less dependent on stricter regulation than more passive companies. Therefore, the adoption of more subtle measures—voluntary actions, eco-audits, eco-labels, and codes of good practice—may be sufficient for pioneering EI companies [23]. However, Els depend more on environmental rules than other innovations [37,54,56].

Finally, other pertinent information concerns the obstacles highlighted by companies. These data are particularly useful to develop, apply, and monitor policies to foster innovation. To assess them, the problems were grouped into three types of barriers: economic (emphasizing the availability and sources of monetary resources), technique (focusing on the direct factors of production and innovation) and institutional (bringing together a variety of actors and forms of interaction) (Figure 5).

The results of barriers to innovation were quite different among industrial activities. In extractive sector, ECOs were shown to be less susceptible to problems, unlike INO (Figure 5a). The exception was in the excessive economic risks item, in the economic barrier. Despite being the most recurrent problem among ECOs, the other two obstacles to the same barrier—high costs of innovation and scarcity of appropriate sources of financing—were not perceived with the same intensity. One speculation is that this phenomenon may be associated with the application of measures to support EIs in extractive activities, such as cheaper access to credit or other financing options, which provide, in return, better environmental performance, which is unknown or even devalued by the sector's



INO. In general, financial barriers (imperfect financial markets, absence of financial assets, scalability, and even weak environmental rules) hinder the development and the diffusion of EIs [28].

Figure 5. Obstacles highlighted by eco-innovative (ECO) and innovative (INO) companies, in the extractive (**a**) and manufacturing (**b**) sectors, according to category, Brazil, 2015–2017. Source: Own elaboration, based on IBGE [44]. Note: data of companies that attributed "high" or "medium" importance to the obstacles.

When examining the manufacturing industry, ECOs indicated the greatest sensitivity to barriers to innovate (Figure 5b). The economic barrier was the most accentuated and the high costs of innovation stood out. In the technical barrier, the most prominent problems were the lack of qualified personnel (also shared by the INO) and the lack of information about technology. This last problem, added to the lack of information on markets and the lack of adequate external technical services, expressed the greatest differences between the two groups of companies. Here, the hypothesis was that ECOs constitute a more specific group of companies, which demand more advanced knowledge and technologies, aimed at a green vanguard [35]. Regarding institutional barriers, the intensity was practically the same in ECO and INO, with a greater signal for the scarce possibilities of cooperation with other companies or institutions in the case of eco-innovators. However, as reported in the literature, in general, the EIs depend more on external sources of knowledge and information than other innovations [54,57].

4.2. Analysis of ECO-Innovation Drivers

Although the literature review exposed a broader set of EI drivers, the investigation analyzed the factors defined by Pintec [45] as follows: voluntary actions (VACT); government support (GOVS); codes of good environmental practices (COGP); market demand (MKTD); high costs of energy, water or raw materials (HCTS); existing environmental standards or taxes on contamination (EEST); environmental standards or future taxes (FEST); company reputation (REPT); public contract requirements (RQPC); and others (OTRS) (Table 1 and Figure 2). The data refer to the extractive and manufacturing industries, as well as the 24 sectors of the latter.

Figure 6 (see interactive heatmap in Figure S1) exposes two clusters: the first, indicated by the column dendrogram (horizontal-axis), groups the EI drivers according to their frequency in the industrial sectors, while the second, line dendrogram (vertical-axis), brings together industrial sectors with a similar distribution of drivers. In both cases, the intensity of the orange color indicates the greater influence of the drive, with the scale in number of standard deviations.



Figure 6. Drivers of eco-innovation on eco-innovative companies by sectors of the extractive and manufacturing industries, Brazil, 2015–2017. Source: Own elaboration, based on IBGE [45].

The analysis of the driver cluster shows distinct scales of value among the drivers of EI. The first group, to the left of the heatmap, places the drivers that contributed least in the decision to develop innovations with environmental gains. Both the OTRS (for which no additional information is given) and the GOVSs had close weights; in turn, the RQPCs were mentioned a slightly more. Government support, subsidies, and other incentives for EIs (technology push) seek to boost the development and application of technologies, processes, products, and organizational procedures that are less aggressive to the environment. Regulatory pull requirements are an environmental policy measure that conditions aspects of the company's ecological conduct to the conclusion of business with the government.

The second group, on the right, was divided into two branches. The first branch grouped EEST and FEST (regulatory pull/push), which presented motivations such as the MKTD driver (market pull). These, in turn, were like those of VACT (market pull), which represent a set of company actions, with the purpose of creating internal practices of good environmental conduct (such as the proper disposal of electronic waste). Here, there is a point to be highlighted regarding the importance given to environmental standards, whether existing or to come. Studies have indicated that EIs are more dependent on regulations than other innovations [37,56–59]. However, in the case of Brazil, this was not verified, either by the environmental norms or by the requirements for public contracts, which are both instruments of environmental policy. The second branch reflects the combination of the three main drivers: REPT and COGP (market pull), and HCTS (costsavings). The reputation is related to the adoption of a green communication strategy, such as the creation of an eco-friendly image, which is also very conducive to stimulate marketing innovations. The codes of good practice comprise tacit measures and agreements that, although not mandatory, act as a type of rule of behavior for companies in each sector, or that are part of a production chain. These drivers showed the highest values. The REPT and COGP are drivers influenced by institutional pressures, which help guide R&D efforts for environmental protection strategies [34]. In turn, the HCTS is a very prominent driver in the literature, indicating that EIs apply far more cost-saving or energy-saving strategies than other innovations [23,30,37,54]. Furthermore, if, on the one hand, firms tend to initiate EI projects in response to market pressures, on the other hand, greater investments in EIs are encouraged by other determinants, such as HCTS [38].

The analysis of the cluster of EI drivers, according to the sector (line dendrogram), allows us to raise, albeit with caution, some hypotheses. It is important to note that the sectorial composition involves differentiated activities included within the same eco-

nomic division; that is, there are different characteristics regarding the intensity of technology/labor, international market orientation [18,60,61], degree of transformation, position in the production chain [39,62], and the size and share capital of the company [18], among other aspects.

The distribution of industrial activities was segmented into two sub-clusters, each formed by two secondary groups, which branched into small groups. In the first subcluster-formed by the first 11 activities arranged in the upper part of the heatmapparticularities were identified that may elucidate the weight of certain EI drivers. An example was the first three sectors: tobacco; paper; and motor vehicles, trailers and bodies. In these activities, EIs were motivated mainly by market pull, REPT, and COGP drivers. Similarly, they are exporting sectors, subject to different types of pressures, such as customers, stakeholders and other actors that relate the activity to others downstream in the production chain (for example, tobacco and forest monocultures). The company's position in the production chain can influence the willingness to eco-innovate, as well as its orientation towards the foreign market [17,18,39]. Another divergent sector was the preparation of leathers and the manufacture of leather goods, travel goods, and footwear, with a strong weight of the driver of voluntary actions. One hypothesis is that, although Pintec does not provide data on the type of product, EIs were potentially of an organizational type, given the relevance of internal environmental protection practices. Based on studies focused on sectors with high polluting potential, it is possible to highlight some points. This is the case for the manufacture of chemicals and energy-intensive industries, which often carry out more EIs due to regulatory pressure or high energy consumption [26,37,42]. Two other activities more sensitive to the regulatory pull/push determinants was the manufacturing of machinery and equipment and the manufacturing of coke, petroleum products, and biofuels.

The second sub-cluster connected 13 activities, plus the extractive and processing industries. In the case of extractive activities, there was a prevalence of existing environmental standards (EEST), (regulatory push/pull), reputation (REPT) and codes of good practice (COGP) (market pull), while in manufacturing, the main drivers were reputation and codes of good practice (both market pull), and the high costs of energy, water, and raw materials (HCTS) (cost-saving). The sector that came closest to the average for manufacturing was metal products. In turn, the activities most influenced by the cost-saving driver were furniture, wearing apparel, printing and reproduction of recordings, and electrical equipment. Strategies to reduce costs and use resources more efficiently are highly significant for eco-innovative companies, even more than for other innovators [29]. The codes of good practice mainly guided EIs in the sectors of rubber and plastics, pharmaceutical products and preparations, and textiles. Finally, basic metals were the most induced to eco-innovate by existing environmental rules.

4.3. Analysis of ECO-Innovation Impacts

Like the analysis of drivers, the examination of the impacts was based on the list of effects on companies (Table 1 and Figure 2), as follows: substitution of materials with others that are less contaminating (SBSM); replacement of energy from renewable fossil fuels (RENG); reduction in the contamination of soil, water, waste, or air (RCON); recycling of waste, wastewater, or materials, for sale or reuse (RECY); and reducing the CO₂ footprint (RCO₂). For each effect, the company indicated the degree of company, being that they were computed only as "high" or "medium" manifestations on impact.

Figure 7 (see interactive heatmap in Figure S2) associates two clusters: one, referred to by the column dendrogram (horizontal-axis), which groups the impacts of EIs according to their occurrence in the industrial sectors, while the other, line dendrogram (vertical-axis), approximates industrial activities with a similar arrangement of impacts. In both clusters, the stronger green color corresponds to the greater relevance of the impact, with the scale being the number of standard deviations.



Figure 7. Impacts of eco-innovation on eco-innovative companies, by sectors of the extractive and manufacturing industries, Brazil, 2015–2017. Source: Own elaboration, based on IBGE [44].

The scale of relevance of environmental impacts perceived by eco-innovative companies was segmented into two sub-clusters. The grouping to the right of the heatmap highlights the most observed impacts. The industrial companies realize that their environmental innovations have made it possible, above all, to recycle waste, including wastewater and materials for reuse (RECY). The reduction in the contamination of soil, water, noise, or air (RCON) was the second most relevant. A common element in the two effects was the similar level of importance across sectors. This is the characteristic that links the two effects on the same branch. At another level was the substitution of raw materials for others that are less harmful to the environment (SBSM). The second sub-cluster brought together the two less relevant effects: the reduction in total CO_2 production (carbon footprint), RCO₂, and the replacement of fossil energies with renewable sources (RENG). Although both impacts are important to mitigate damage to the environment, this drew attention to the low level of energy change, by green technologies for energy generation. Furthermore, it is pertinent to mention that EIs that reduce CO_2 emissions are highly dependent on subsidies, basic research, and public technology programs [8].

The examination of the impacts by sector of activity indicated a very distinct cut, formed by a large group with three branches and a group of only one activity—in this case, that of wood products. This isolated position of the sector is due to the high frequency of impact in four of the five predicted. In addition, this activity showed the greatest manifestation regarding energy transformation, using cleaner sources of energy generation (RENG). In general, the other sectors had between two or three more important impacts. Regardless of this provision, it is important to reinforce the reservations regarding the characteristics of each sector, which make it possible to establish only a few deductions. One of them was the lesser relevance of the effect of substituting materials for others that are less harmful to the environment (SBSM) in some sectors, such as wood products; coke and refined petroleum products; food and beverage products; basic metals; and extractive activities. All these activities depend heavily on specific materials for which there are no direct substitutes. This is the case for water for the beverage and food industry; wood for products of this nature; coke, oil, and sugar cane (or other materials used for this purpose, such as soy, corn, and biomass) for fuels; and minerals for extraction activities.

4.4. Drivers, Impacts and Innovation Variables: Correlated Effects

To complete the heatmap analysis, we calculated the correlations among the drivers (10), impacts (5), and other selected innovation variables (91) captured by Pintec (Appendix A). The purpose of this exercise was to establish a profile of eco-innovators, which could guide actions to encourage EI. The results of this exercise are shown in Figure 8.

VADIADIEC	DRIVERS								IMPACTS					
VARIABLES	EEST	FEST	GOVS	MKTD	REPT	VACT	COGP	HCTS	RQPC	SBSM	RENG	RCON	RECY	RCO ₂
Innovative intensity rate	No	No	No	No	No	No	No	No	No	0.521 *	No	No	No	No
Main responsible product innovation - the company		-0.750 **	No	No	No	No	No	No	-0.549 *	No	No	No	0.545 *	No
Main responsible product innovation - another company and institute	No	0.601 *	No	No	No	No	No	No	0.652 **	No	No	No	No	No
Main responsible process innovation - the company	No	-0.565 *	No	No	No	No	No	No	No	No	No	No	No	No
Main responsible process innovation - another company and institute	No	-0.527 *	No	No	No	No	No	No	No	No	No	No	No	No
Benefits of innovation - maintenance of market share	No	No	No	No	No	No	No	No	No	No	No	No	No	0.525 *
Benefits of innovation - expansion of market share	No	No	No	No	No	No	No	No	No	0.525 *	No	No	No	No
Benefits of innovation - reduced labor costs	No	No	0.509 *	No	No	No	No	0.681 **	No	No	No	No	No	No
Benefits of innovation - reduction of energy consumption	No	No	0.554 *	No	No	No	No	0.649 **	0.567 *	No	No	No	No	No
Benefits of innovation - expansion of the control of aspects related to health and safety	No	No	No	No	No	No	No	No	No	No	No	0.647 **	No	No
Benefits of innovation - compliance with standard regulations and standards	No	No	No	No	No	No	No	No	No	No	No	No	0.610 *	No
Barriers of innovation - scarcity of adequate external technical services	No	No	No	No	No	No	No	No	0.520 *	No	No	No	No	No
Information sources - another company in the group	No	No	No	No	No	No	0.560 *	No	No	No	No	No	No	No
Information sources - professional training centers and technical assistance	No	No	No	No	No	No	No	0.528 *	No	No	No	No	No	No
ooperation - suppliers	No	No	No	No	No	0.530 *	No	No	No	No	No	No	No	No
Organizational Innovation - management technique	No	No	No	No	No	No	No	No	No	0.510 *	No	No	No	No
Organizational Innovation - environmental management technique	No	No	No	0.528 *	No	No	No	No	No	No	No	No	No	No
Eco-innovators that produced renewable energy	No	No	No	No	No	No	No	No	No	No	0.564 *	No	No	No
EEST	No	0.604 *	No	No	No	No	No	No	No	No	No	No	No	No
FEST	0.604 *	No	No	No	No	No	No	No	No	No	No	No	-0.606 *	No
REPT	No	No	No	No	No	No	0.828 **	No	No	No	No	No	No	No
COGP	No	No	No	No	0.828 **	No	No	No	No	No	No	No	No	No
HCTS	No	No	No	No	No	No	No	No	0.616 *	No	No	No	No	No
RQPC	No	No	No	No	No	No	No	0.616 *	No	No	No	No	No	No
RCON	No	No	No	No	No	No	No	No	No	No	No	No	No	0.519 *
RECY	No	-0.606 *	No	No	No	No	No	No	No	No	No	No	No	0.560 *
RCO ₂	No	No	No	No	No	No	No	No	No	No	No	0.519 *	0.560 *	No

Figure 8. Correlation coefficient—selected variables. Note: No—Not significant; * significant at 1%; ** significant at 0.1%.

Among the aspects observed, one was that eco-innovators motivated by REPT were also influenced by COBP, both drivers of market pull. In EEST and FEST, and in the HCTS (of energy, water, or raw materials) and RQPC, the same happened. In the latter—drivers of a different nature—it is likely that when the company eco-innovates looking for cost-saving, it is also more attentive to other market opportunities, such as meeting environmental conditions to supply products or services to the public sector.

In terms of impacts, the reduction in CO_2 emissions (RCO₂) was linked to two other effects: the RCON (grouping soil, water, waste, or air) and the RECY (contemplating waste, wastewater, or materials). ECO-innovators, who showed a decrease in their carbon footprint, also exhibited a reduction in contamination, in addition to the effects of recycling.

In RECY, there was an inverse correlation with FEST. This result was interesting and corroborates with previous studies. As already noted, recycling is generally associated with the use of more widespread and less complex technologies [37,63]. In this sense, when an eco-innovative company is guided by environmental standards or future taxes, it expects regulations to become more restrictive, and will probably prefer to develop innovations with greater effect in reducing contamination, which recycling practices are more limited in achieving.

The correlations with other innovation variables highlighted pertinent elements. In drivers, eco-innovators induced by FEST normally have another company and institute as the main responsible for product innovation. This relationship reinforces the fact that eco-innovators are more dependent on external agents, especially when there is an expectation of greater rigidity of environmental standards [37,42,57]. In contrast, these standards or taxes were inversely related when the main actor responsible for product innovation is the company. In process EIs, regardless of the main innovation agent, the relationship with

FEST remains the opposite. In other words, companies guided by these standards tend not to eco-innovate in the process.

Government encouragement (GOVS) was associated with reduced labor and energy costs. Moreover, those motivated by market demand (MKTD) tended to adopt environmental management techniques. Recalling that these techniques involve, among other aspects, an environmental communication policy aimed mainly at answering questions from consumers and stakeholders about the company's environmental behavior, this result reinforces the importance of environmental management systems for EIs [34]. In turn, companies that were guided by voluntary efforts (VACT) tended to cooperate with suppliers. A similar observation was made between the COGP and the relevance of another company in the group in the sources of information. In these two situations, there was a greater willingness to establish links downstream of the production chain or with companies in the same sector. It was interesting to note that the high cost driver (HCTS) was related to the benefits of reducing labor and energy costs, and to sources of information with professional training and technical assistance centers. Regarding RQPC, these are linked to the condition that the main executor of the product EI is another company and institute, the benefits of decreased energy consumption, the absence of adequate external technical services, and conversely, when the main executor of product innovation is the company.

In respect of impacts, the replacement of materials (SBSM) was associated with the rate of innovative intensity of the companies, their expansion in the market and the application of management techniques. The substitution of energy (RENG) was related to the production of renewable energy. The RCON tended to increase the control of aspects related to health and safety. The recycling effect was mainly observed when it led to the framing of regulations and standards, and when the main developer of the product innovation was another company and institute. Finally, eco-innovators who saw a reduction in their CO_2 footprint more directly perceived the company's maintenance gains in the market.

5. Discussion

To promote the development of EIs, some aspects must be considered. The first is the fact that they need much more stimulation than other innovations, mainly regarding access to external sources of knowledge and information, the granting of subsidies and the support of public technology programs [8,54]. In the case of Brazilian eco-innovative companies, the scarce possibilities for cooperation with other companies or institutions, the lack of information about technology and the market, the scarcity of external technical services and the high costs of innovation were recurrent impediments in manufacturing.

Although this relevance has not been observed in extractive activities, an addressing investment to promote EIs would be very auspicious. In developing countries such as Brazil, the support of universities and research centers, especially in the environmental and technological areas, can simultaneously help to overcome economic and environmental problems, and can provide opportunities for EIs [37]. As the study indicated, the proximity to other external agents proved to be relevant for eco-innovators motivated by the reduction in high costs. A similar observation was made in the relationship between voluntary actions and cooperation with suppliers.

However, the effectiveness of actions such as this depends greatly on the degree of articulation and commitment between the actors involved. The government has a key role in this process, mainly the public institutions responsible for industrial, innovation and environmental policies [15]. As highlighted in the study, even with the encouragement of some actions, there are important corrections to be made. A good example is the eco-innovators who, even induced by the requirements of public contracts, face the lack of adequate external technical services.

The synergy between the instruments of these policies can increase environmental engagement in the economy, involving companies and consumers, the public and private sectors, universities and research centers, institutional and civil society representatives. The

goals set out in SDG 9 are linked to this environmental commitment. The modernization of infrastructures and the improvement of the industry to make them sustainable, the increase in scientific research and technological capabilities of the industrial sectors, and the encouragement of innovation are goals that are part of the same transformation circuit.

From the perspective of companies, they need to develop skills in the scope of sustainability, improving products, processes, and organizational and management forms. For example, initiatives related to organizational innovations have indicated interesting links with the drivers of market demand and the exchange of materials for others polluting less.

Despite being an internal task that must be undertaken in the long term [22], the benefits achieved by EIs are an incentive both for eco-innovators to advance in new projects and for other companies to realize the advantages of eco-innovating, valuing in its innovation strategy the reduction in environmental impacts. As the study revealed, although EI requires greater effort on the part of eco-innovators—as indicated in the index of in-house R&D activities—they perceived with greater emphasis the gains in the three categories analyzed, competition, production, and protection.

The analysis of drivers and the impacts of EIs provided interesting insights. Examining the drivers, the conclusion was that EIs in Brazil still receive scarce attention from the public sphere. This was evident in the results of government support drivers (technology push), requirements for public contracts (regulatory pull), and environmental standards, existing or expected (regulatory pull/push), which were less relevant than the other inducers. This result is in line with other studies, such as Mazzucato and Penna [16], who pointed out a lack of cohesiveness in the Brazilian innovation system, as we have seen at the beginning of this article, namely through aspects that show the difficulties in creating innovation policy instruments aimed at environmental innovation, in dialogue with industrial and environmental policies. Here, public financial support is essential to encourage EIs. On the other hand, the greater representativeness of the company's reputation drivers and codes of good practice (both market pull), and high cost of materials (cost-saving), signals that the development of EIs is highly associated with a market response. The analysis allowed us to make some hypotheses based on the formation of sub-clusters, such as the connection of certain drivers with the sector's level of exposure in the foreign market, or the position sector in a broader production chain.

The examination of the impacts resulting from EIs was equally interesting, either by complementing the analysis of the inductors, or by indicating the types of EI results most observed in the country. The most relevant effects were the recycling of waste, the reduction in contamination, and the replacement of materials with others that are less harmful to the environment. In contrast, the least verified were the reduction in carbon footprint and the replacement of non-renewable sources of energy with cleaner sources.

Although Pintec does not inform the purpose of the EI, it is desirable to establish some considerations. Among them is the small relevance of adopting clean energy. The transition to a more sustainable model is strongly linked to this substitution, as stated in SDG 7, which aims to increase the share of renewable energy by 2030. Efforts to reduce energy consumption require more advanced technologies and higher R&D expenditures. In Brazil, an interesting initiative is the adoption of a solar energy system for public lighting, an EI implemented by the public sector [63]. For this reason, the development and use of clean energy occurs mainly in central countries, while less technologically demanding options, such as recycling, predominate among semi-peripheral and peripheral countries [19]. This last situation was verified in Brazil. In EIs, the effects associated with more widespread and less complex technologies, such as the recycling of waste, wastewater, or materials, prevailed. This consideration can be extended to the effects of reducing the contamination of soil, water, air and noise, which tend to represent a first stage of EI [39], associated with the use of end-of-pipe, which seeks to improve the quality levels of these elements, through the application of filters or other devices for this purpose. It is important to note that EIs are most often associated with cleaner production measures, unlike end-of-pipe technologies, that reduce environmental damage without changing the production process [64].

Focusing on the sector, a feature of semi-peripheral countries such as Brazil is the high representativeness of sectors of medium or low technological intensity, many of which also have high levels of polluting potential. Despite sharing these characteristics, this does not mean that they are homogeneous; quite the contrary, there is a variety of aspects that differentiate them—market orientation, technological intensity, labor intensity and qualification, and production chain, among others. Hence, the technological options and EI opportunities will differ from sector to sector. Thus, it is presumed that an available part of the stock of clean technologies belongs to a given sector. Therefore, the characteristics of each production activity guides both existing technological opportunities and influences their degree of innovativeness and, likewise, affect the chances that EIs are developed and adopted [39]. This is a key point to be considered in the measures and actions to promote EIs.

Here, it is interesting to note the relationship, or lack thereof, between the sectors with the same profile in drivers and EI impacts. In groups of companies with the same profile of drivers and impacts, it was not possible to detect similarities in technological terms. Particularly in the drivers, it was expected that the sectors of the same group had the condition of related variety [65], i.e., the firms of different sectors had some cognitive proximity and, consequently, intersectoral complementarity that could explain the similar innovative profile. However, the possibility must be raised that this diversity is simply random and, therefore, there is no justification for this. ECO-innovative companies follow an "ecological" technological trajectory for specific reasons, shared in their sector, and which are not present in units from other sectors of the same profile defined by heatmaps.

6. Conclusions

It is increasingly urgent to establish a new model of production and consumption that is sustainable. An alternative to achieve this objective and advance in the development of a sustainable technological model is to encourage EIs. However, semi-peripheral countries such as Brazil will have to deal with their technological limits to access this new green model. Given this scenario, the study set out to establish an overview of EI in the country, with the aim of bringing elements to expand the dialogue among innovation, industry and environment. The study indicated that eco-innovators represent a third of all innovators. As highlighted, these companies have shown greater effort to innovate and greater susceptibility to the benefits and obstacles of innovation; their eco-innovation strategy are mainly driven by market factors; and the impacts are mainly associated with the use of more widespread and less complex technologies.

At the global level, numerous ways to encourage more sustainable production and consumption are discussed. One way is to rethink the economic incentives for EI. An interesting measure is the implementation of industrial certifications and credits that guarantee that the production processes comply with environmental requirements [15].

Semi-peripheral countries such as Brazil can overcome the technological gap by relying on emulation from mission-oriented innovation policies [66] that, for example, can promote an assignment, such as diminishing carbon emissions in a specific period, or a transformative innovation policy [67] that highlights an integrated and systemic approach, considering the negative social and environmental effects of innovation in its framework. Both cases highlight the promotion of investments in innovation (especially in EIs) with the leading role in this process [68].

These types of policy can be carried out through various instruments, which can be on the supply side, such as tax incentives for R&D, or on the demand side, such as public procurement policies, or even by both orientations (supply and demand), such as technical standards [69].

However, the analysis undertaken here may highlight some interesting preferential paths. Returning to the results of EI drivers, among the most valued inductors were reputation and codes of good practice. Both inductors—with a strong correlation as indicated by the study—are tacit and, therefore, not mandatory. Thus, two policy instruments seem

ideal for raising awareness and impelling (or compelling) more companies to innovate in environmental terms. Exercising influence on both the supply and demand sides, these instruments can be used for technological forecasting and regulation.

Technological forecasting aims to sensitize and alert public and private managers about trends in technological trajectories at national and global levels and, at the same time, about the benefits of considering these trends. As previously mentioned, the theme of EIs or sustainable technological development is in the list of global megatrends, highlighting the strong link between technological changes and the environmental dimension [5,7,10,11,70]. It is important to emphasize that this instrument also intends to provoke improvements in the discourse in general, which is fundamental for changing an outdated technological trajectory. On the other hand, the regulation instrument—also correlated—has an important role in the same direction as technological forecasting, but on a compulsory basis. In short, regulations seek to increase the demand for (environmental) innovations and improve the overall technological structure.

In addition to this pair of policy instruments, many others would apply. Among those that should be highlighted are those linked to the promotion of interaction and learning at regional and national scales [69]. Directed to the offer, three instruments are highlighted here: agglomeration policies, support for collaboration, and innovation networks. All of these emphasize the importance of the agents' interdependence to innovate in new and disruptive themes to reach more advanced levels of general technological development. For example, it is worth mentioning the case of eco-industrial clusters, which can be fostered by policies, where the waste from one plant is used as input by another plant, etc. [71], forming clusters of sustainable industries, whose production is associated with the circular economy. The EIs represent a type of transformation process, capable of creating a socioeconomic system based on the idea of the circular economy [72]. In this sense, it is very pertinent to examine whether innovative institutional contexts—regional and national—encourage environmental technological change and, therefore, the EIs [73].

Another interesting perspective is to analyze the phenomenon of the development and diffusion of EIs from the territory [74,75]. Although (eco)innovation requires an analytical knowledge base [76], which results in closer links with universities and the formation of complex international knowledge exchange networks [77], it also requires a synthetic knowledge base [76], which is linked to knowledge and skills based on local, regional and national experience [77].

As a research agenda, we consider that this study can stimulate further analysis. The expectation is that the presented sectorial portrait will serve as an initial stage to investigate groups of sectors that share similarities, such as high polluting potential or belonging to certain production chains. Here, case studies can provide valuable information, particularly to guide eco-innovation policies. Another further research agenda is the analysis of related variety, or systems of related firms in eco-innovation issues from a regional development perspective [65], for example: occupational structures as information networks and also through the flow of labor among industries; the technological relatedness among industries, a fundamental issue for developing countries [78]; or research relationship among individuals or scientific institutions. In all these cases, the relatedness analysis typically looks for the probability that a region establishes an economic activity within the scope of eco-innovation as a function of related activities in that place.

Finally, we believe that EIs have the intrinsic characteristic of avoiding one of the main negative effects of innovations in general, one of the components of the "dark side of innovation" [79], namely, that of environmental degradation. Indeed, from the Industrial Revolution to the present day, many innovations have contributed to the environmental changes that are currently threatening the planet through pollution, loss of biodiversity, climate change, etc. Therefore, in view of a necessary and urgent sustainable global development, EIs are essential in all productive activities.

Supplementary Materials: The following are available online at https://www.mdpi.com/article/10.3390/su13148065/s1, Figure S1: Drivers of eco-innovation on eco-innovative companies, by sectors of the extractive and manufacturing industries, Brazil, 2015–2017; and Figure S2: Impacts of eco-innovation on eco-innovative companies, by sectors of the extractive and manufacturing industries, Brazil, 2015–2017.

Author Contributions: Conceptualization, F.Q.S. and I.G.P.T.; methodology, F.Q.S. and I.G.P.T.; validation, F.Q.S. and I.G.P.T.; formal analysis, F.Q.S. and I.G.P.T.; investigation, F.Q.S. and I.G.P.T.; visualization, F.Q.S. and I.G.P.T.; writing—original draft preparation, F.Q.S. and I.G.P.T.; writing—review and editing, F.Q.S. and I.G.P.T. All authors have read and agreed to the published version of the manuscript.

Funding: This research received support from the Centre of Studies in Geography and Spatial Planning (CEGOT), funded by national funds through the Foundation for Science and Technology (FCT) under the reference UIDB/04084/2020.

Acknowledgments: The authors would like to thank the special tabulations provided by the Pintec/IBGE team, as well as the participants of the study presentation session (in V Meeting of Industrial Economics and Innovation) and the reviewers for their comments and suggestions.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

List of variables from the correlation analysis.

- 1. Innovation rate
- 2. Innovative intensity rate
- 3. Innovative intensity rate R&D internal
- 4. Product innovation rate—national market
- 5. Product innovation rate—world market
- 6. Process innovation rate—national market
- 7. Process innovation rate—world market
- 8. Main responsible product innovation—the company
- 9. Main responsible product innovation—another company in the same group
- 10. Main responsible product innovation—in cooperation with another company
- 11. Main responsible product innovation—another company and institute
- 12. Main responsible process innovation—the company
- 13. Main responsible process innovation—another company in the same group
- 14. Main responsible process innovation—in cooperation with another company
- 15. Main responsible process innovation—another company and institute
- 16. Number of workers exclusively dedicated to R&D
- 17. Number of researchers in R&D activities
- 18. Number of graduate workers in R&D activities
- 19. Number of undergraduate workers in R&D activities
- 20. Expenditure on innovative activities—internal R&D activities
- 21. Expenditure on innovative activities—external R&D activities
- 22. Expenditure on innovative activities—external knowledge
- 23. Expenditure on innovative activities—software
- 24. Expenditure on innovative activities—machinery and equipment
- 25. Expenditure on innovative activities-training
- 26. Expenditure on innovative activities—market innovation
- 27. Expenditure on innovative activities—industrial and other projects
- 28. Innovative activities—Internal R&D (high or medium relevance)
- 29. Innovative activities-external acquisition of R&D*
- 30. Innovative activities—acquisition of another knowledge *
- 31. Innovative activities—acquisition of software *
- 32. Innovative activities—acquisition of machinery and equipment *
- 33. Innovative activities—training *

- 34. Innovative activities—introduction of new technical innovations *
- 35. Innovative activities—industrial design and other techniques *
- 36. Benefits of innovation—improving the quality of goods and services *
- 37. Benefits of innovation—expansion of the range of goods and services *
- 38. Benefits of innovation-maintenance of market share *
- 39. Benefits of innovation-expansion of market share *
- 40. Benefits of innovation—opening new markets *
- 41. Benefits of innovation—increase in productive capacity *
- 42. Benefits of innovation—increased production flexibility *
- 43. Benefits of innovation-reduction of production costs *
- 44. Benefits of innovation-reduced labor costs *
- 45. Benefits of innovation-reduction of raw material consumption *
- 46. Benefits of innovation—reduction of energy consumption *
- 47. Benefits of innovation—reduction of water consumption *
- 48. Benefits of innovation—expansion of the control of aspects related to health and safety *
- 49. Benefits of innovation-compliance with standard regulations and standards *
- 50. Barriers of innovation—excessive economic risks *
- 51. Barriers of innovation—high costs of innovation *
- 52. Barriers of innovation—lack of adequate sources of financing *
- 53. Barriers of innovation—organizational rigidity *
- 54. Barriers of innovation—lack of qualified personnel *
- 55. Barriers of innovation—lack of information about technology *
- 56. Barriers of innovation—lack of market information *
- 57. Barriers of innovation-difficulty adapting to standards, rules and regulations *
- 58. Barriers of innovation—scarcity of adequate external technical services *
- 59. Barriers of innovation—centralization of innovation activity in another group company *
- 60. Barriers of innovation—few possibilities for cooperation with other companies/ institutions *
- 61. Barriers of innovation—weak consumer response to new products *
- 62. Information sources—R&D Department *
- 63. Information sources—other areas in the company *
- 64. Information sources—another company in the group *
- 65. Information sources—suppliers *
- 66. Information sources—customers or consumers *
- 67. Information sources—competitors *
- 68. Information sources—independent consultants and consultants *
- 69. Information sources—universities or other higher education centers *
- 70. Information sources—research institutes or technology centers *
- 71. Information sources—professional training centers and technical assistance *
- 72. Information sources-testing, assay and certification institutions *
- 73. Information sources—conferences, meetings and specialized publications *
- 74. Information sources—trade fairs and exhibitions *
- 75. Information sources-computerized information networks *
- 76. Cooperation—customers and consumers *
- 77. Cooperation—suppliers *
- 78. Cooperation—competitors *
- 79. Cooperation—another company in the group *
- 80. Cooperation—consultancy firms *
- 81. Cooperation—universities and research institutes *
- 82. Cooperation—professional training centers and technical assistance *
- 83. Cooperation—testing, testing and certification institutions *
- 84. Organizational Innovation—management technique

- 85. Organizational Innovation—environmental management technique
- 86. Organizational Innovation—work organization
- 87. Organizational Innovation—external relations
- 88. Marketing Innovation-concepts/strategies
- 89. Marketing Innovation—aesthetics, design and others
- 90. ECO-innovators that publish Sustainability Reports
- 91. ECO-innovators that produced renewable energy* high or medium relevance.

References

- 1. United Nations. The Sustainable Development Goals. 2018. Available online: https://sdgs.un.org/goals (accessed on 17 December 2019).
- Kemp, R.; Pearson, P. *Final Report MEI Project about Measuring ECO-Innovation*; UM-Merit: Maastricht, The Netherlands, 2007. Available online: https://www.oecd.org/env/consumption-innovation/43960830.pdf (accessed on 9 October 2016).
- 3. Arundel, A.; Kemp, R. Measuring ECO-Innovation. Working Paper Series; United Nations University, 2009. Available online: http://collections.unu.edu/eserv/unu:324/wp2009-017.pdf (accessed on 16 September 2016).
- 4. Kemp, R. ECO-innovation: Definition, Measurement and Open Research Issues. Econ. Politica 2010, 27, 397–420. [CrossRef]
- Mazzucato, M.; Semieniuk, G.; Watson, J. What Will It Take to Get Us to a Green Revolution? Policy Paper 2015, U. Sussex. Available online: https://www.sussex.ac.uk/webteam/gateway/file.php?name=what-will-it-take-to-get-us-a-green-revolution. pdf&site=264 (accessed on 16 January 2016).
- Perez, C.; Leach, T. A Smart Green 'European Way of Life': The Path for Growth, Jobs and Wellbeing. Beyond the Technological Revolution, Working Paper Series. 2018. Available online: http://beyondthetechrevolution.com/wp-content/uploads/2014/10/ BTTR_WP_2018-1.pdf (accessed on 9 August 2018).
- 7. Lema, R.; Fu, X.; Rabellotti, R. Green windows of opportunity: Latecomer development in the age of transformation toward sustainability. *Ind. Corp. Chang.* 2021, 29, 1193–1209. [CrossRef]
- 8. Horbach, J.; Rammer, C.; Rennings, K. Determinants of eco-innovations by type of environmental impact—The role of regulatory push/pull, technology push and market pull. *Ecol. Econ.* **2012**, *78*, 112–122. [CrossRef]
- 9. Freeman, C. The greening of technology and models of innovation. *Technol. Forecast. Soc. Chang.* **1996**, *53*, 27–39. [CrossRef]
- 10. Mazzucato, M.; Semieniuk, G. Public financing of innovation: New questions. Oxf. Rev. Econ. Policy 2017, 33, 24–48. [CrossRef]
- Perez, C. Unleashing a golden age after the financial collapse: Drawing lessons from history. *Environ. Innov. Soc. Transit.* 2013, 6, 9–23. [CrossRef]
- 12. Perez, C. Cambio tecnológico y oportunidades de desarrollo como blanco móvil. *Rev. CEPAL* **2001**, *75*, 115–136. Available online: http://www.eclac.org/publicaciones/xml/3/19323/perez.pdf (accessed on 7 April 2013). [CrossRef]
- Perez, C. Innovation Systems and Policy: Not only for the Rich? Working Papers in Technology Governance and Economic Dynamics 2012, n. 42. pp. 1–23. Available online: http://technologygovernance.eu/files/main/2012071005451212.pdf (accessed on 13 April 2013).
- 14. Kemp, R.; Never, B. Green transitions, industrial policy, and economic development. *Oxf. Rev. Econ. Pol.* **2017**, 33, 66–84. [CrossRef]
- Altenburg, T.; Rodrik, D. Green industrial policy: Accelerating structural change towards wealthy green economies. In *Green Industrial Policy. Concept, Policies, Country Experiences*; Altenburg, T., Assamann, C., Eds.; UN Environment, German Development Institute: Bonn, Germany, 2017; pp. 1–20. Available online: https://www.un-page.org/files/public/green_industrial_policy_book_aw_web.pdf (accessed on 27 February 2020).
- Mazzucato, M.; Penna, C. *The Brazilian Innovation System: A Mission-Oriented Policy Proposal*; Centro de Gestão e Estudos Estratégicos (CGEE): Brasília, Brazil, 2016. Available online: https://www.cgee.org.br/documents/10195/909424/The_Brazilian_ Innovation_System-CGEE-MazzucatoandPenna-FullReport.pdf. (accessed on 8 April 2016).
- 17. Horbach, J. Determinants of environmental innovation—New evidence from German panel data sources. *Res. Policy* **2008**, *37*, 163–173. [CrossRef]
- Triguero, Á.; Cuerva, M.C.; Álvarez-Aledo, C. Environmental Innovation and Employment: Drivers and Synergies. Sustainability 2017, 9, 2057. [CrossRef]
- 19. Horbach, J. Empirical determinants of eco-innovation in European countries using the community innovation survey. *Environ. Innov. Soc. Transit.* **2016**, *19*, 1–14. [CrossRef]
- 20. Porter, M.E.; Linde, C. Toward a New Conception of the Environment-Competitiveness Relationship. *J. Econ. Perspect.* **1995**, *9*, 97–118. Available online: http://www.jstor.org/stable/2138392 (accessed on 5 February 2013). [CrossRef]
- 21. Freeman, C.; Soete, L. The Economics of Industrial Innovation, 3rd ed.; Routledge: London, UK, 2004.
- 22. Bossle, M.B.; de Barcellos, M.D.; Vieira, L.M.; Sauvée, L. The drivers for adoption of eco-innovation. *J. Clean. Prod.* 2016, 113, 861–872. [CrossRef]
- 23. Rennings, K. Redefining innovation—ECO-innovation research and the contribution from ecological economics. *Ecol. Econ.* **2000**, 32, 319–332. [CrossRef]

- 24. Jaffe, A.B.; Newell, R.G.; Stavins, R.N. A tale of two market failures: Technology and environmental policy. *Ecol. Econ.* 2005, 54, 164–174. [CrossRef]
- González-Benito, J.; González-Benito, Ó. Environmental proactivity and business performance: An empirical analysis. *Omega* 2005, 33, 1–15. [CrossRef]
- Díaz-García, C.; Moreno, A.G.; Martínez, F.J.S. Eco-innovation: Insights from a literature review. *Innovation* 2015, 17, 6–23. [CrossRef]
- 27. Ryszko, A. Proactive environmental strategy, technological eco-innovation and firm performance-case of Poland. *Sustainability* **2016**, *8*, 156. [CrossRef]
- Ruiz, J.D.G.; Botero-Botero, S.; Duque-Grisales, E. Financial ECO-Innovation as a Mechanism for Fostering the Development of Sustainable Infrastructure Systems. *Sustainability* 2018, 10, 4463. [CrossRef]
- 29. Barbieri, N.; Ghisetti, C.; Gilli, M.; Marin, G.; Nicolli, F. A survey of the literature on environmental innovation based on main path analysis. *J. Econ. Surv.* 2016, 33, 596–623. [CrossRef]
- Triguero, A.; Moreno-Mondéjar, L.; Davia, M.A. Drivers of different types of eco-innovation in European SMEs. *Ecol. Econ.* 2013, 92, 25–33. [CrossRef]
- 31. Hojnik, J.; Ruzzier, M. What drives eco-innovation? A review of an emerging literature. *Environ. Innov. Soc. Transit.* 2016, 19, 31–41. [CrossRef]
- 32. Del Río, P.; Peñasco, C.; Romero-Jordán, D. Distinctive Features of Environmental Innovators: An Econometric Analysis. *Bus. Strat. Environ.* 2015, 24, 361–385. [CrossRef]
- 33. Dosi, G. Technical Change and Industrial Transformation: The Theory and an Application to the Semiconductor Industry; Macmillan Press: London, UK, 1984; p. 338.
- 34. Kesidou, E.; Demirel, P. On the drivers of eco-innovations: Empirical evidence from the UK. *Res. Policy* 2012, 41, 862–870. [CrossRef]
- 35. De Marchi, V. Environmental innovation and R&D cooperation: Empirical evidence from Spanish manufacturing firms. *Res. Policy* **2012**, *41*, 614–623. [CrossRef]
- 36. Pereira, R.M.; MacLennan, M.L.F.; Tiago, E.F. Interorganizational cooperation and eco-innovation: A literature review. *Int. J. Innov. Sci.* 2020, 12, 477–493. [CrossRef]
- 37. Horbach, J. Do eco-innovations need specific regional characteristics? An econometric analysis for Germany. *Rev. Reg. Res.* 2014, 34, 23–38. [CrossRef]
- Paraschiv, D.M.; Nemoianu, E.L.; Langa, C.A.; Szabó, T. Eco-innovation, responsible leadership and organizational change for corporate sustainability. *Amf. Econ.* 2012, 32, 404–419. Available online: https://www.amfiteatrueconomic.ro/ArticolEN.aspx? CodArticol=1133 (accessed on 25 January 2018).
- 39. Carrillo-Hermosilla, J.; Del Río, P.; Könnölä, T. *Eco-Innovation: When Sustainability and Competitiveness Shake Hands*, 1st ed.; Pal-Grave Macmillan: London, UK, 2009.
- 40. Wagner, M. Empirical influence of environmental management on innovation: Evidence from Europe. *Ecol. Econ.* **2008**, *66*, 392–402. [CrossRef]
- 41. Berrone, P.; Fosfuri, A.; Gelabert, L.; Gomez-Mejia, L.R. Necessity as the mother of 'green' inventions: Institutional pressures and environmental innovations. *Strat. Manag. J.* **2013**, *34*, 891–909. [CrossRef]
- 42. Del Río, P.; Peñasco, C.; Romero-Jordán, D. What drives eco-innovators? A critical review of the empirical literature based on econometric methods. *J. Clean. Prod.* 2016, 112, 2158–2170. [CrossRef]
- 43. Kemp, R.; Pontoglio, S. The innovation effects of environmental policy instruments—A typical case of the blind men and the elephant? *Ecol. Econ.* **2011**, *72*, 28–36. [CrossRef]
- 44. Instituto Brasileiro de Geografia e Estatística. Pesquisa de Inovação 2017: Notas Técnicas; IBGE: Rio de Janeiro, Brazil, 2020.
- 45. Instituto Brasileiro de Geografia e Estatística. Pesquisa de Inovação 2017: Tabulação Especial; IBGE: Rio de Janeiro, Brazil, 2020.
- 46. Conceicao, P.; Heitor, M.V.; Vieira, P.S. Are environmental concerns drivers of innovation? Interpreting Portuguese innovation data to foster environmental foresight. *Technol. Forecast. Soc. Chang.* **2006**, *73*, 266–276. [CrossRef]
- 47. Queiroz, J.M.; Podcameni, M.G.V.B. Estratégia inovativa das firmas brasileiras: Convergência ou divergência com as questões ambientais? *Rev. Bras. Inovação* 2013, *13*, 187–224. [CrossRef]
- 48. Chapman, K. Environmental policy and industrial location in the United States. In *Institutions and Geographical Patterns*; Flowerdew, R., Ed.; Croom Helm: London, UK, 1982; pp. 141–168.
- Galili, T.; O'Callaghan, A.; Sidi, J.; Sievert, C. heatmaply: An R package for creating interactive cluster heatmaps for online publishing. *Bioinformatics* 2018, 34, 1600–1602. [CrossRef] [PubMed]
- 50. Rstudioteam. *RStudio: Integrated Development for R [Computer Software]*; RStudio: Boston, MA, USA, 2020. Available online: https://rstudio.com/ (accessed on 17 July 2020).
- 51. Storme, T.; Derudder, B.; Dörry, S. Introducing cluster heatmaps to explore city/firm interactions in world cities. *Comput. Environ. Urban. Syst.* **2019**, *76*, 57–68. [CrossRef]
- 52. Organization for Economic Co-Operation and Development (OECD). *Oslo Manual: Guidelines for Collecting and Interpreting Innovation Data*, 3rd ed.; OECD Publishing: Paris, France, 2005; p. 166. [CrossRef]
- 53. Leitão, J.; De Brito, S.; Cubico, S. Eco-Innovation Influencers: Unveiling the Role of Lean Management Principles Adoption. *Sustainability* **2019**, *11*, 2225. [CrossRef]

- Horbach, J.; Oltra, V.; Belin, J. Determinants and Specificities of ECO-Innovations Compared to Other Innovations—An Econometric Analysis for the French and German Industry Based on the Community Innovation Survey. *Ind. Innov.* 2013, 20, 523–543. [CrossRef]
- 55. Hájek, P.; Stejskal, J. R&D Cooperation and Knowledge Spillover Effects for Sustainable Business Innovation in the Chemical Industry. *Sustainability* **2018**, *10*, 1064. [CrossRef]
- Carrillo-Hermosilla, J.; del Río, P.; Könnölä, T. Diversity of eco-innovations: Reflections from selected case studies. J. Clean. Prod. 2010, 18, 1073–1083. [CrossRef]
- 57. Leitão, J.; Pereira, D.; De Brito, S. Inbound and Outbound Practices of Open Innovation and ECO-Innovation: Contrasting Bioeconomy and Non-Bioeconomy Firms. *J. Open Innov. Technol. Mark. Complex.* **2020**, *6*, 145. [CrossRef]
- 58. Oltra, V. Environmental Innovation and Industrial Dynamics: The Contributions of Evolutionary Economics. W. P. GREThA. 2008. Available online: https://core.ac.uk/download/pdf/6297007.pdf (accessed on 27 October 2017).
- 59. Ghisetti, C.; Pontoni, F. Investigating policy and R&D effects on environmental innovation: A meta-analysis. *Ecol. Econ.* **2015**, *118*, 57–66. [CrossRef]
- 60. Horbach, J.; Jacob, J. The relevance of personal characteristics and gender diversity for (eco-)innovation activities at the firm-level: Results from a linked employer-employee database in Germany. *Bus. Strat. Environ.* **2018**, 27, 924–934. [CrossRef]
- 61. Choi, H.; Yi, D. Environmental innovation inertia: Analyzing the business circumstances for environmental process and product innovations. *Bus. Strat. Environ.* **2018**, *27*, 1623–1634. [CrossRef]
- 62. Galera-Quiles, M.; Piedra-Muñoz, L.; Galdeano-Gómez, E.; Carreño-Ortega, A. A Review of ECO-Innovations and Exports Interrelationship, with Special Reference to International Agrifood Supply Chains. *Sustainability* **2021**, *13*, 1378. [CrossRef]
- 63. Koeller, P.; Miranda, P.; Lustosa, M.; Podcameni, M. *Ecoinovação: Revisitando o Conceito;* Texto para Discussão 2556; IPEA: Rio de Janeiro, Brasil, 2020; p. 60. Available online: http://repositorio.ipea.gov.br/bitstream/11058/9960/1/td_2556.pdf (accessed on 22 October 2020).
- 64. Frondel, M.; Horbach, J.; Rennings, K. End-of-pipe or cleaner production? An empirical comparison of environmental innovation decisions across OECD countries. *Bus. Strat. Environ.* 2007, *16*, 571–584. [CrossRef]
- 65. Asheim, B.T.; Boschma, R.; Cooke, P. Constructing Regional Advantage: Platform Policies Based on Related Variety and Differentiated Knowledge Bases. *Reg. Stud.* 2011, 45, 893–904. [CrossRef]
- Mazzucato, M. Mission-Oriented Innovation Policy: Challenges and Opportunities; Working Paper; UCL—Inst. for Innov. and Public Purpose: London, UK, 2017. Available online: https://www.ucl.ac.uk/bartlett/public-purpose/sites/public-purpose/files/ moip-challenges-and-opportunities-working-paper-2017-1.pdf (accessed on 27 April 2019).
- 67. Schot, J.; Steinmueller, W.E. Three frames for innovation policy: R&D, systems of innovation and transformative change. *Res. Policy* **2018**, 47, 1554–1567. [CrossRef]
- Zagato, L. Ainda é possível que os países em desenvolvimento façam seu catching up no século XXI? *Braz. J. Political Econ.* 2019, 39, 527–543. [CrossRef]
- 69. Edler, J.; Fagerberg, J. Innovation policy: What, why, and how. Oxf. Rev. Econ. Policy 2017, 33, 2–23. [CrossRef]
- Schot, J.; Kanger, L. Deep transitions: Emergence, acceleration, stabilization and directionality. *Res. Policy* 2018, 47, 1045–1059. [CrossRef]
- 71. Aoyama, Y.; Murphy, J.T.; Hanson, S. Key Concepts in Economic Geography; SAGE Publications: London, UK, 2011.
- 72. De Jesus, A.; Mendonça, S. Lost in Transition? Drivers and Barriers in the ECO-innovation Road to the Circular Economy. *Ecol. Econ.* **2018**, *145*, 75–89. [CrossRef]
- 73. González, P.D.R. The empirical analysis of the determinants for environmental technological change: A research agenda. *Ecol. Econ.* **2009**, *68*, 861–878. [CrossRef]
- 74. Tartaruga, I.; Sperotto, F. Rethinking Clusters in the Sense of Innovation, Inclusion, and Green Growth. In *Rethinking Clusters*; Sedita, S., Blasi, S., Eds.; Sustainable Development Goals Series; Springer: Cham, Switzerland, 2021; pp. 101–110. [CrossRef]
- 75. Podcameni, M.G.; Cassiolato, J.E.; Lustosa, M.C.; Marcellino, I.; Rocha, P. Exploring the convergence between sustainability and local innovation systems from a southern perspective: What Brazilian empirical evidence has to offer. *Local Econ. J. Local Econ. Policy Unit* **2019**, *34*, 825–837. [CrossRef]
- 76. Asheim, B. Differentiated knowledge bases and varieties of regional innovation systems. Innovation 2007, 20, 223–241. [CrossRef]
- 77. Binz, C.; Truffer, B. Global Innovation Systems—A conceptual framework for innovation dynamics in transnational contexts. *Res. Policy* **2017**, *46*, 1284–1298. [CrossRef]
- 78. Hidalgo, C.; Balland, P.A.; Boschma, R.; Delgado, M.; Feldman, M.; Frenken, K.; Glaeser, E.; He, C.; Kogler, D.; Morrison, A.; et al. The Principle of Relatedness. In Proceedings of the Unifying Themes in Complex. Systems IX, ICCS 2018, Springer Proceedings in Complexity, Cambridge, MA, USA, 22–27 July 2018; Morales, A., Braha, D., Minai, A., Bar-Yam, Y., Eds.; Springer: Cham, Switzerland, 2018. [CrossRef]
- 79. Coad, A.; Nightingale, P.; Stilgoe, J.; Vezzani, A. Editorial: The dark side of innovation. Ind. Innov. 2021, 28, 102–112. [CrossRef]